Author's response to reviews

Title: Upper arm metastasis of an Oral Squamous Cell Carcinoma

Authors:

Matthias C Wurm (Matthias.Wurm@uk-erlangen.de)
Emeka Nkenke (Emeka.Nkenke.extern@uk-erlangen.de)
Friedrich W Neukam (Friedrich.Neukam@uk-erlangen.de)
Tobias Möst (Tobias.Moest@uk-erlangen.de)
Rainer Lutz (Rainer.Lutz@uk-erlangen.de)
Falk Wehrhan (Falk.Weehrhan@uk-erlangen.de)
Kathrin Brunner (Kathrin.Brunner@uk-erlangen.de)
Konstantinos T Mitsimponas (Konstantinos.Mitsimponas@uk-erlangen.de)
Philipp Schlechtweg (Philipp.Schlechtweg@uk-erlangen.de)
Cornelius von Wilmowsky (Cornelius.Wilmowsky@uk-erlangen.de)

Version: 2
Date: 15 October 2014

Author’s response to reviews:

Dear Mrs. Nolasco,

please find enclosed our revised manuscript. The following changes were made:
- Title Page now contains email-addresses
- Consent uploaded
- Line numbering included
- changed Disclosure Statement to Competing interest.
- included list of abbreviations.
- abstract structured into sections
- consent section included
- authors contribution included
- acknowledgement included
- care checklist uploaded

I appreciate your help.

Yours sincerely,
Matthias Wurm